Author's response to reviews

Title: The efficacy of risk scores for predicting abdominal wound dehiscence: a case-controlled validation study.

Authors:

Jakub Kenig (jkenig@cm-uj.krakow.pl)
Piotr Richter (jkenig@cm-uj.krakow.pl)
Anna Lasek (aniad303@gmail.com)
Katarzyna Zbierska (alistra16@gmail.com)
Sabina Zurawska (s.zurawska@gmail.com)

Version: 5  Date: 18 April 2014

Author's response to reviews:

Editorial Office, BMC Surgery

We did not change anything in the text since our study was revised for the first time, addressing all minor and major comments of the reviewers.

After the paper was transferred to BMC Surgery, we reformatted the text according to the BMS’s manuscript guidelines and also added (according to the editorial comment) the statement about institutional ethics committee.

That is why we submit once again our manuscript highlighting only the statement about institutional ethics committee in the material and method section.

Sincerely,
Jakub Kenig